This article was downloaded by: [18.7.29.240] On: 18 April 2015, At: 12:48 Publisher: Institute for Operations Research and the Management Sciences (INFORMS) INFORMS is located in Maryland, USA



Management Science

Publication details, including instructions for authors and subscription information: http://pubsonline.informs.org

Does Corporate Social Responsibility Lead to Superior Financial Performance? A Regression Discontinuity Approach

Caroline Flammer

To cite this article:

Caroline Flammer (2015) Does Corporate Social Responsibility Lead to Superior Financial Performance? A Regression Discontinuity Approach. Management Science

Published online in Articles in Advance 19 Feb 2015

. http://dx.doi.org/10.1287/mnsc.2014.2038

Full terms and conditions of use: <u>http://pubsonline.informs.org/page/terms-and-conditions</u>

This article may be used only for the purposes of research, teaching, and/or private study. Commercial use or systematic downloading (by robots or other automatic processes) is prohibited without explicit Publisher approval, unless otherwise noted. For more information, contact permissions@informs.org.

The Publisher does not warrant or guarantee the article's accuracy, completeness, merchantability, fitness for a particular purpose, or non-infringement. Descriptions of, or references to, products or publications, or inclusion of an advertisement in this article, neither constitutes nor implies a guarantee, endorsement, or support of claims made of that product, publication, or service.

Copyright © 2015, INFORMS

Please scroll down for article-it is on subsequent pages



INFORMS is the largest professional society in the world for professionals in the fields of operations research, management science, and analytics.

For more information on INFORMS, its publications, membership, or meetings visit <u>http://www.informs.org</u>

http://dx.doi.org/10.1287/mnsc.2014.2038 © 2015 INFORMS

Does Corporate Social Responsibility Lead to Superior Financial Performance? A Regression Discontinuity Approach

Caroline Flammer

Ivey Business School, University of Western Ontario, London, Ontario N6G 0N1, Canada, cflammer@ivey.uwo.ca

This study examines the effect of shareholder proposals related to corporate social responsibility (CSR) on financial performance. Specifically, I focus on CSR proposals that pass or fail by a small margin of votes. The passage of such "close call" proposals is akin to a random assignment of CSR to companies and hence provides a quasi-experiment to study the effect of CSR on performance. I find that the adoption of close call CSR proposals leads to positive announcement returns and superior accounting performance, implying that these proposals are value enhancing. When I examine the channels through which companies benefit from CSR, I find that labor productivity and sales growth increase after the vote. Finally, I document that close call CSR proposals differ from non-close proposals along several dimensions. Accordingly, although my results imply that adopting close call CSR proposals is beneficial to companies, they do not necessarily imply that CSR proposals are beneficial in general.

Data, as supplemental material, are available at http://dx.doi.org/10.1287/mnsc.2014.2038.

Keywords: corporate social responsibility; financial performance; regression discontinuity; shareholder proposals

History: Received January 4, 2013; accepted July 11, 2014, by Wei Jiang, finance. Published online in *Articles in Advance*.

1. Introduction

Does corporate social responsibility (CSR) lead to superior corporate financial performance (CFP)? Anecdotal evidence points toward a positive relationship between the two. For instance, the UK retailer Marks & Spencer implemented an ambitious CSR program in 2007 "with the ultimate goal of becoming the world's most sustainable major retailer" (Brokaw 2012). Five years later, this program turned out to be very profitable. In particular, chief executive officer (CEO) Marc Bolland qualifies the outcome as "a strong business case for sustainability, with £185 million in net benefits" (Brokaw 2012). More generally, recent surveys indicate that a large majority of CEOs believe that CSR can improve a firm's competitiveness and is critical to its future success (see, e.g., Lacy et al. 2010, Haanaes et al. 2012).

Understanding the relationship between CSR and CFP has spurred a large academic literature. In their review, Margolis et al. (2007) report that 167 studies have examined the CSR–CFP link between 1972 and 2007. These studies have been surveyed in no fewer than 16 review articles. The typical approach in this literature is to regress measures of CFP (Tobin's *Q*, return on assets, etc.) on measures of CSR (e.g., the

Kinder, Lydenberg, and Domini index of social performance). In their meta-analysis of these studies, Margolis et al. conclude that the overall correlation between CSR and CFP—more precisely, the coefficient of CSR in the above regression—is positive but small.

A limitation of this literature is that CSR is endogenous with respect to CFP, i.e., a company's decision to engage in CSR activities likely correlates with unobservable firm characteristics that may also affect CFP. For example, it could be that companies engage in CSR *because* they are more profitable or expect their future profitability to be higher. Or it could be that CEOs who implement long-term CSR strategies are also those who are more likely to perform well (e.g., since they are more talented). In sum, the positive correlation between CSR and CFP that emerges from the literature, albeit interesting, does not warrant a causal interpretation.¹

Going beyond such a correlation is difficult. From an empirical perspective, the ideal experiment would be to randomly assign firms into a "high CSR group"

¹Similarly, Margolis et al. (2007) conclude their meta-analysis by highlighting the need to move beyond the "simple correlation between CSP and CFP" (p. 33).

and a "low CSR group" and compare their financial performance following this "treatment." Obviously, such an ideal experiment would be difficult and unreasonably costly to implement in the field.

In this paper, I consider a quasi-natural experiment that is very close in spirit to this ideal experiment. Specifically, I compare the effect of shareholdersponsored CSR proposals that pass or fail by a small margin of votes in annual meetings. The passage of such "close call" proposals is akin to a random assignment of CSR to companies and hence is uncorrelated with firm characteristics. Intuitively, there is no reason to expect any systematic difference between a company for which a CSR proposal passes with 50.1% of the votes and a company for which a similar proposal fails with 49.9% of the votes. Accordingly, close call CSR proposals provide a source of random variation in CSR that can be used to estimate the causal effect of CSR on CFP. The general approach of comparing outcomes just above and below a discontinuous threshold is known as *regression discontinuity design* (RDD) in the economic literature. In this paper, the discontinuity arises because, around the 50% majority threshold, a minor difference in vote shares leads to a discrete change (i.e., a discontinuity) in the adoption of CSR policies.²

The data on CSR proposals are obtained from Risk-Metrics and SharkRepellent, which compile information on shareholder proposals of U.S. publicly traded companies that came to a vote from 1997 to 2012. The proposals are classified into two broad categories: social issues (e.g., the implementation of nondiscrimination policies) and environmental issues (e.g., the reduction of CO_2 emissions).³

My main finding is that the passage of close call CSR proposals significantly increases shareholder value. On the day of the shareholder meeting, a CSR proposal that passes by a narrow margin of votes yields an abnormal return of 0.92% compared with a CSR proposal that fails marginally. Since shareholder proposals are not binding, this estimate measures only the effect of approving a CSR proposal as opposed to the effect of implementing a CSR proposal. The latter can be approximated by rescaling the estimated coefficient by the probability of implementing the proposal. I find that CSR proposals have a 52% probability of being implemented. Accordingly, my results imply that implementing a close call CSR proposal leads to an increase in shareholder value by about 1.77%. This finding is consistent with the view that CSR is a valuable resource, which is in line with, e.g., the resource-based view of the firm (e.g., Hart 1995, Russo and Fouts 1997), instrumental stakeholder theory (e.g., Jones 1995), and Porter and Kramer's (2006, 2011) shared value argument.

I then examine whether the increase in shareholder value depends on the level of CSR prior to the vote. I find that the value gains are larger for companies with relatively low levels of CSR. This suggests that CSR is a resource with decreasing marginal returns; i.e., the CSR–CFP relationship is concave. Intuitively, initial efforts to improve CSR may yield substantial benefits (the "low-hanging fruits" of CSR). However, as companies keep increasing their social performance, the returns from an additional CSR initiative may decrease.

Next, I examine the channels through which CSR increases shareholder value. I find that the passing of close call CSR proposals has a positive impact on operating performance in the years following the vote. When I further examine what explains the increase in operating performance, I find that the adoption of close call CSR proposals leads to an increase in labor productivity and sales growth. This evidence suggests that these proposals improve job satisfaction and help companies cater to customers that are responsive to sustainable practices.

Although my results imply that close call CSR proposals are beneficial to companies, they do not necessarily imply that CSR proposals are beneficial in general. The vast majority of CSR proposals receive little support at shareholder meetings (the average vote outcome is merely 13.5%), suggesting that shareholders may not find them desirable. Hence, it seems plausible that close call proposals may not be representative of the average CSR proposal. When I characterize the subset of close call proposals, I indeed find that they differ from non-close proposals along several dimensions. First, they are more likely to address employee satisfaction and the mitigation of environmental hazards. Second, a textual analysis of their support statement shows that they more frequently contain arguments linking CSR to performance. Third, close call proposals are more frequently found among companies operating in "stakeholder-sensitive" industries, i.e., industries in which performance depends greatly on the relationship with employees and customers. Overall, this suggests that close call CSR proposals are more likely to be related to performance in some way. Accordingly, one must be cautious in extrapolating my results to non-close proposals.

The remainder of this paper is organized as follows. Section 2 describes the competing hypotheses

² See Lee and Lemieux (2010) and Roberts and Whited (2013) for surveys of RDD applications in the economics and finance literature. For a formal treatment of the RDD methodology as well as practical advice, see Imbens and Lemieux (2008).

³ Some scholars use a broader definition of CSR that also encompasses corporate governance—i.e., "ESG" (environmental, social, and governance). The definition of CSR used in this paper does not include corporate governance. For a study of governance proposals, see Cuñat et al. (2012).

3

on the CSR–CFP relationship. Section 3 describes the data and empirical methodology. Section 4 presents the results. Section 5 discusses the external validity. Section 6 concludes.

2. Background: The Relationship Between Social and Financial Performance

The relationship between social and financial performance has received considerable attention in the theory literature. The early literature, in the spirit of shareholder theory (e.g., Friedman 1962, 1970), views social responsibility as a "donation" from shareholders to stakeholders that reduces profits. Similarly, CSR may be the outcome of an agency conflict between shareholders and managers (Jensen and Meckling 1976): companies' social engagement may be driven by managers' own social preferences or their desire to establish overly friendly relationships with specific stakeholders.

Whereas the early literature predicts a negative relationship between CSR and financial performance, subsequent research emphasizes the potential value of CSR. For example, Freeman's (1984) stakeholder theory argues that companies should consider the interests of everyone who can substantially affect, or be affected by, the welfare of the company. This theory has been extended in various ways (for a review, see Agle et al. 2008). For instance, instrumental stakeholder theory argues that CSR efforts are actions taken to benefit stakeholders with the ultimate goal of benefiting shareholders; i.e., CSR is "instrumental" to firm performance (e.g., Jones 1995). Similarly, in line with the literature on sustainability in business and the resource-based view of the firm, companies may engage in CSR to improve their efficiency and enhance, e.g., their reputation, brand, and trust (e.g., Barney 1991; Hart 1995; Porter 1991; Porter and Kramer 2006, 2011; Russo and Fouts 1997). In turn, such actions may attract new customers (socially conscious customers, "green" consumers, etc.), increase the companies' profitability, and enhance their competitiveness (e.g., Flammer 2014).

A large set of anecdotal evidence suggests that a growing number of multinational companies including, e.g., General Electric (GE), Google, IBM, Intel, Johnson & Johnson, Marks & Spencer, Nestle, Unilever, and Walmart—see the benefits of creating "shared value" (Porter and Kramer 2011) and, in particular, expect to gain a competitive advantage from CSR initiatives. For example, GE's CEO Jeffrey Immelt, discussing GE's "ecomagination" program, stated, "We did it from a business standpoint from Day 1, [...], it was never about corporate social responsibility" (Lohr 2011). In sum, both the recent literature and anecdotal evidence suggest that CSR may be a *causal* determinant of financial performance. In this study, I examine whether such a causal link is supported by the data.

3. Data and Methodology

3.1. Shareholder Proposals

The data on shareholder proposals are obtained from two databases: RiskMetrics and SharkRepellent. Risk-Metrics covers shareholder proposals that came to a vote from 1997 to 2011 at S&P 1500 companies as well as approximately 400–500 additional widely held companies. SharkRepellent's proxy voting database includes shareholder proposals from a broad universe of about 4,000 companies in the Russell 3000 index from 2005 to 2012. Both databases include firm identifiers, a description of the proposal, the date of the annual meeting, the proposal's sponsor, the voting requirement, and the outcome of the vote.

I merge both databases to obtain a comprehensive data set of shareholder proposals that came to a vote between 1997 and 2012. I then restrict the sample to shareholder proposals that are related to CSR. In RiskMetrics, such proposals are identified by the resolution type "SRI" (social responsible initiative); in SharkRepellent, they are identified by the proposal category "Social/Environmental Issues." The final sample consists of 2,729 CSR proposals. Two specific examples are provided in Table 1.

Table 1	Examples	of CSR	Proposals

Panel A: Example of CSR proposal that was closely rejected

Company:	Lear Corporation
Meeting date: Proposal:	May 11, 2006 "[T]he shareholders request that the company commit
FTOPOSal.	itself to the implementation of a code of conduct
	based on the aforementioned ILO human rights standards and United Nations' Norms on the
	Responsibilities of Transnational Corporations with
	Regard to Human Rights, by its international suppliers and in its own international production facilities, and commit to a program of outside, independent
	monitoring of compliance with these standards."
Voting result:	Rejected (49.8% of the votes)
Source:	SharkRepellent
Panel B:	Example of CSR proposal that was closely approved
Company:	HCC Insurance Holdings, Inc.
Meeting date:	May 10, 2007
Proposal:	"The Shareholders request that management implement equal employment opportunity policies based on the aforementioned principles prohibiting discrimination based on sexual orientation and gender identity."
Voting result:	Passed (52.2% of the votes)
Source:	SharkRepellent

Figure 1 Distribution of Votes for Shareholder CSR Proposals



Notes. This figure presents the histogram of the vote shares in favor of the CSR proposals. The horizontal axis indicates the vote share in 5% intervals. The vertical axis indicates the frequency of proposals. The sample consists of all CSR proposals in the RiskMetrics and SharkRepellent databases from 1997 to 2012.

CSR proposals typically do not fare well at annual meetings. Figure 1 provides the histogram of the vote outcome of the 2,729 CSR proposals. As is shown, most proposals receive very little support: about 75% of the proposals receive less than 20% of favorable votes. Only a small fraction of proposals make it close to the majority threshold. This pattern suggests that most CSR proposals may be "symbolic" in nature. Shareholders submit them not so much because they expect the proposals to pass but rather to bring social issues to the attention of management and the public (Loss and Seligman 2004).

My identification strategy relies on proposals with a close call outcome. Fortunately, although the number of close call proposals is small relative to the total number of proposals, it is sufficiently large in absolute terms: 61 proposals received a vote share within the $\pm 5\%$ interval around the majority threshold and 122 within the $\pm 10\%$ interval. This lends sufficient power to the identification.

Table 2 provides more details about the 2,729 CSR proposals. Panel A reports the frequency of shareholder proposals by year as well as summary statistics on the vote outcomes. As can be seen, both the number of proposals and the proportion of favorable votes have increased over time. In particular, although the average percentage of votes in favor was merely 9% in the first half of the sample (1997–2004), it increased to 17% in the second half (2005–2012). This evolution is consistent with previous evidence documenting an increase in shareholders' awareness for CSR issues over time (e.g., Flammer 2013).

Panel B in Table 2 further reports the breakdown of the proposals according to the different types of CSR. The classification is obtained from the SharkRepellent database, where CSR proposals are partitioned into two broad categories and nine subcategories. The precise definition of each subcategory is provided in the appendix. RiskMetrics does not provide a classification of CSR proposals but does include a onesentence description of the proposal's content. I use this description to manually assign each proposal to a SharkRepellent category. As can be seen, the proposals that are most likely to be approved are those pertaining to labor issues (5.27% are approved).

Panel C in Table 2 further provides a breakdown according to the type of proposal sponsor. The most common sponsors are religious groups, yet they are also the least likely to succeed. The most successful activists are public pension funds and SRI funds, who see 3.89% and 3.36%, respectively, of their proposals being approved.

3.2. Methodology

Cuñat et al. (2012; henceforth CGG) develop a methodology that adapts the regression discontinuity framework so as to estimate the effect of shareholder proposals on shareholder returns and other outcome variables. This paper uses the same methodology, except that CSR proposals are used instead of governance proposals. In the following, I provide a brief

Table 2 Shareholder CSR Proposals

		Panel A	A: Summary statisti	cs by year			
Year	Shareholder proposals	Approved proposals	Approved proposals (%)	Average vote outcome (%)	SD vote outcome (%)	Vote outcome $\pm 5\%$	Vote outcome $\pm 10\%$
1997	111	0	0.00	7.05	3.72	0	0
1998	119	0	0.00	7.83	5.39	0	0
1999	126	1	0.79	7.82	7.61	0	0
2000	144	0	0.00	7.42	4.75	0	0
2001	159	1	0.63	9.23	7.60	0	0
2002	162	2	1.23	10.01	8.77	1	2
2003	142	1	0.70	11.84	11.13	0	2
2004	187	4	2.14	11.36	13.10	0	1
2005	195	2	1.03	9.97	9.14	0	3
2006	206	5	2.43	14.08	12.98	5	7
2007	215	7	3.26	17.23	15.73	15	20
2008	212	6	2.83	15.32	14.58	8	12
2009	196	8	4.08	18.01	14.88	10	19
2010	197	4	2.03	18.66	14.77	6	20
2011	179	7	3.91	20.15	16.33	8	19
2012	179	3	1.68	19.69	14.26	8	17
Total	2,729	51	1.87	13.48	12.97	61	122
	F	Panel B: Summ	ary statistics by typ	e of CSR proposa	al		
Proposal type	Shareholder proposals	Approved proposals	Approved proposals (%)	Average vote outcome (%)	SD vote outcome (%)	Vote outcome $\pm 5\%$	Vote outcome $\pm 10\%$
Environment issues	648	8	1.23	15.13	13.16	17	38
Environmental issues	504	5	0.99	13.08	11.58	7	19
Sustainability report	144	3	2.08	22.29	15.67	10	19
Social issues	2,081	43	2.07	12.97	12.86	44	84
Add minorities/women to board	79	2	2.53	18.29	12.60	2	4
Animal rights	130	0	0.00	4.99	3.19	0	0
Health issues	391	1	0.26	7.54	7.60	0	1
Human rights	227	1	0.44	11.90	11.12	0	4
Labor issues	455	24	5.27	16.42	14.69	23	36
Other social issues related	355	9	2.54	9.29	11.64	7	9
Political issues	444	6	1.35	19.11	13.84	12	30
		Panel C:	Summary statistics	by sponsor			
	Shareholder	Approved	Approved	Average vote	SD vote	Vote outcome	Vote outcome
Proposal sponsor	proposals	proposals	proposals (%)	outcome (%)	outcome (%)	$\pm 5\%$	±10%
Individual	449	6	1.34	8.88	9.77	5	7
Public pension fund	437	17	3.89	21.28	14.71	27	49
Religious	834	5	0.60	10.55	10.31	5	14
SRI fund	506	17	3.36	16.94	15.16	13	33
							10
Union	201	4	1.99	15.48	12.15	7	13

Notes. This table displays the frequency of shareholder proposals, the frequency and percentage of approved proposals, the mean and standard deviation of the percentage of favorable votes, and the frequency of proposals whose vote share is within 5% and 10% of the majority threshold. The sample consists of all CSR proposals in the RiskMetrics and SharkRepellent databases from 1997 to 2012. In panel A, proposals are classified by year; in panel B, by type of CSR; and in panel C, by the proposal's sponsor.

summary of this methodology. For more details, see Section II of CGG.

3.2.1. Regression Discontinuity in Shareholder Votes. The objective is to estimate the effect of passing a CSR proposal on an outcome variable for firm *i* at time *t*, denoted by y_{it} (e.g., the stock market reaction on the day of the shareholder meeting).

The proposal receives a vote share v_{it} . Whether the proposal is approved is denoted by the indicator variable $pass_{it} = 1(v_{it} \ge v^*)$, where v^* is the majority threshold.⁴ To estimate the effect of CSR proposals on y_{it} , I would ideally need a randomized assignment

⁴ For most companies, the threshold is 50% of the votes. In the rare cases when companies have a stricter majority requirement

6

of the *pass* indicator. The RDD is helpful in approximating this ideal setting, since it relies on proposals that pass or fail by a narrow margin of votes. Arguably, whether a proposal passes by 50.1% of the votes or whether it fails by 49.9% is as good as random. Accordingly, such close call CSR proposals provide a source of random variation in the adoption of CSR proposals that can be used to estimate the causal effect of passing a CSR proposal on y_{it} .

The RDD can be implemented by estimating the difference in average y_{it} between CSR proposals that pass or fail by a small margin of votes. Although this difference does provide an unbiased estimate of the effect of passing a CSR proposal on y_{it} , it comes at the cost of discarding all non-close proposals. A more efficient estimate can be obtained by using all CSR proposals and approximating the continuous relationship between y_{it} and v_{it} with a polynomial in v_{it} , allowing for a discontinuous jump at the majority threshold v^* . Following CGG, I allow for a different polynomial for observations on the left-hand side of the threshold $P_l(v_{it}, \gamma_l)$ and on the right-hand side of the threshold $P_r(v_{it}, \gamma_r)$. The RDD specification can be written as follows:

$$y_{it} = \beta \times pass_{it} + P_l(v_{it}, \gamma_l) + P_r(v_{it}, \gamma_r) + \varepsilon_{it}.$$
 (1)

The estimate of β captures the discontinuity at the majority threshold and hence provides a consistent estimate of the causal effect of passing a CSR proposal on y_{it} .⁵ To account for within-firm dependence across observations, I cluster standard errors at the firm level. Throughout the paper, I use polynomials of order 3. The results are similar if I use second- or fourth-order polynomials instead.

3.2.2. Multiple Periods and Multiple Votes. As CGG emphasize, the specification in Equation (1) is subject to two potential caveats. First, the shareholder vote at time t may have an impact on outcomes at t+1, t+2, etc. Second, for each firm and meeting date, shareholders may have to vote on more than one CSR proposal.

To address these two caveats, CGG propose a multiperiod version of specification (1) in which multiple CSR proposals in a given meeting are "aggregated." In a nutshell, this procedure can be implemented as follows. First, the multiple periods are accounted for by using a panel data set in which, for each firmmeeting (i,t), observations at time $t + \tau$ are pooled for multiple τ , including $\tau < 0$. Specifically, I use observations in periods t - 2 to t + T (as in CGG, T is up to seven days for abnormal returns and up to four years for annual variables such as the return on assets). The coefficient on the dummy for whether the proposal has passed β_{τ} is then τ -specific and is constrained to zero for $\tau < 0$. Similarly, the parameters of the polynomials $\gamma_{l,\tau}$ and $\gamma_{r,\tau}$ are allowed to vary for $\tau \geq 0$. Since observations before and after the event are pooled together, an advantage of this specification is that I can include firm-meeting fixed effects in the regression α_{it} . Doing so accounts for any unobservable firm characteristics that are constant during the event window. Following CGG, I further include fixed effects for the time period relative to the meeting date α_{τ} ("distance-to-the-election" fixed effects) and fixed effects for the calendar year α_c .

Second, CGG propose an aggregation procedure that accounts for the possibility of multiple proposals in the same shareholder meeting. Implementing this procedure is straightforward. Essentially, the *pass* dummy in Equation (1) is replaced by the sum of the *pass* dummies for all *n* proposals that are voted on in the same meeting (i.e., the "aggregated *pass* dummy" becomes a count variable for the number of CSR proposals that are passed in the meeting). The polynomials in vote share are adjusted accordingly. A formal treatment of this aggregation procedure is provided in CGG.

The multiperiod version of Equation (1) that accounts for the possibility of multiple proposals can be expressed as follows:

$$y_{i,t+\tau} = \beta_{\tau} \sum_{k=1}^{n} pass_{it}^{k} + \left[P_l \left(\sum_{k=1}^{n} v_{it}^{k}, \gamma_{l,\tau}^{k} \right) + P_r \left(\sum_{k=1}^{n} v_{it}^{k}, \gamma_{r,\tau}^{k} \right) \right] + \alpha_{it} + \alpha_{\tau} + \alpha_{c} + \varepsilon_{i,t+\tau}.$$
(2)

This specification is essentially Equation (7) in CGG (p. 1958). As in specification (1), standard errors are clustered at the firm level and third-order polynomials are used throughout.

3.3. Variable Definitions and Sample Characteristics

3.3.1. Abnormal Returns. The main dependent variable used in this paper is the abnormal return on the day of the shareholder meeting (t = 0). Using abnormal returns is appealing because they provide an estimate of the effect of passing a CSR proposal on firm value, thus capturing all potential channels through which CSR may benefit shareholders. Another advantage of using stock returns is the high signal-to-noise ratio: the stock market reaction on the day of the vote is most likely attributable to the vote itself. In contrast, accounting measures of performance—e.g., the return on assets in the following year—may capture other events that have occurred during the year.

⁽e.g., a "supermajority" requirement of two-thirds of the votes), I adjust v^* accordingly.

⁵ Since shareholder proposals are not binding, β only measures the effect of approving a CSR proposal as opposed to the effect of implementing a CSR proposal. The latter can be approximated by rescaling the estimate of β by the probability of implementation; see §4.2.

Following CGG, I compute abnormal returns using the four-factor model of Carhart (1997). The four factors are the market return (the return on the market portfolio minus the risk-free rate), the size factor ("small minus big"), the book-to-market factor ("high minus low"), and the momentum factor ("up minus down"). In robustness checks, I show that my results are similar if I use the market model instead.

Daily stock return data are obtained from the Center for Research in Security Prices. The four factors are obtained from Kenneth French's website (http://mba .tuck.dartmouth.edu/pages/faculty/ken.french/data _library.html, accessed January 31, 2015). The coefficients of the four-factor model are estimated by ordinary least squares using an estimation period of 200 trading days that starts 20 trading days prior to the shareholder meeting. To be included in the sample, a stock needs to have at least 15 days with nonmissing returns during the 200-day estimation period.

3.3.2. Summary Statistics. The 2,729 CSR proposals used in this study correspond to 1,845 firm-year observations. Table 3 provides summary statistics for the abnormal return on the day of the shareholder meeting (first row) as well as several other characteristics, which are constructed from various data sources.

The companies' financials are computed from Standard & Poor's Compustat in the fiscal year that ends prior to the shareholder meeting. Market value is the number of shares outstanding multiplied by the stock price at the end of the fiscal year. Total assets is the book value of assets. Return on assets (ROA) is the ratio of operating income before depreciation to the book value of assets. Return on equity (ROE) and net profit margin (NPM) are defined similarly except that the denominator is the book value of equity plus deferred taxes and investment tax credit for ROE and sales for NPM. Tobin's Q is the ratio of the market value of total assets (book value of assets plus the

Table 3 **Summary Statistics**

Characteristic	N	Mean	Median	SD	10th %ile	90th %ile
Abnormal return on meeting day	1,845	0.001	0.001	0.023	-0.019	0.024
Market value (\$ million)	1,845	37,881	12,673	62,748	1,286	108,424
Total assets (\$ million)	1,838	77,365	16,539	238,890	1,441	138,354
ROA	1,810	0.137	0.137	0.083	0.037	0.251
ROE	1,561	0.362	0.333	0.270	0.161	0.721
NPM	1,810	0.194	0.173	0.140	0.065	0.377
Tobin's Q	1,588	1.890	1.493	1.102	0.985	3.500
KLD index	1,687	4.06	3.00	3.82	0.00	9.00
G-index	1,666	9.23	9.00	2.62	6.00	13.00
Institutional ownership (%)	1,750	63.27	68.40	26.99	8.38	91.34
Inside ownership (%)	1,761	0.64	0.00	4.14	0.00	0.97
Labor productivity	1,824	437	332	328	128	1,090
Capital expenditures	1,788	0.051	0.042	0.042	0.005	0.106
Sales growth	1,781	0.071	0.062	0.132	-0.102	0.255
Leverage	1,836	0.264	0.254	0.158	0.049	0.474
Cash	1,833	0.101	0.057	0.120	0.008	0.252
Labor intensity (industry-level)	1,845	0.321	0.282	0.183	0.126	0.586
B2C industry	1,845	0.425	0.000	0.495	0.000	1.000

Notes. The 2,729 CSR proposals considered in this study correspond to 1,845 firm-year observations. Abnormal returns on the day of the vote are computed using the four-factor model of Carhart (1997). All Compustat variables are computed in the fiscal year that ends prior to the date of the vote. Market value is the number of shares outstanding (Compustat item "CSHO") multiplied by the stock price at the end of the fiscal year ("PRCC_F"). Total assets is the book value of total assets ("AT"). ROA (return on assets) is the ratio of operating income before depreciation ("OIBDP") to the book value of total assets. ROE (return on equity) and NPM (net profit margin) are defined similarly except that the denominator for ROE is the book value of equity ("CEQ") plus deferred taxes and investment tax credit ("TXDITC") and for NPM is sales ("SALE"). Tobin's Q is the ratio of the market value of total assets to the book value of total assets. The market value of total assets is the book value of total assets plus the market value of equity (CSHO \times PRCC_F) minus the sum of the book value of equity (CEQ) plus deferred taxes and investment tax credit (TXDITC). Labor productivity is the ratio of sales (SALE) to the number of employees ("EMP"). Capital expenditures is the ratio of capital expenditures ("CAPX") to total assets. Sales growth is the growth in sales (SALE) compared with the previous fiscal year. Leverage is the ratio of debt in current liabilities ("DLC") and long-term debt ("DLTT") to total assets. Cash is the ratio of cash and short-term investments ("CHE") to total assets. The KLD index is the number of CSR strengths (from the Kinder, Lydenberg, and Domini database). The G-index is the governance index of Gompers et al. (2003), obtained from RiskMetrics. The KLD index and G-index are measured in the calendar year prior to the vote. Institutional ownership is the percentage of shares owned by institutional investors in the quarter that ends prior to the date of the vote (from the Thomson-Reuters Institutional Holdings database). Inside ownership is the percentage of shares owned by all executives of the firm in the fiscal year prior to the date of the vote (from ExecuComp). Labor intensity is the median ratio of the sum of staff expense ("XLR") and pension and retirement expense ("XPR") to sales (SALE) in the company's two-digit SIC industry in the year preceding the vote. B2C industry is a dummy variable equal to 1 if the company operates in the B2C sector according to the classification of Lev et al. (2010, p. 188) based on four-digit SIC codes. All ratios are winsorized at the 5th and 95th percentiles of their empirical distribution. Note that the number of observations may change as a result of missing values. %ile, percentile.

market value of equity minus the sum of the book value of equity plus deferred taxes and investment tax credit) to the book value of assets. *Labor productivity* is the ratio of sales to the number of employees. *Capital expenditures* is the ratio of capital expenditures to total assets. *Sales growth* is the growth in sales compared with the previous fiscal year. *Leverage* is the ratio of debt in current liabilities and long-term debt to total assets. Finally, *cash* is the ratio of cash and short-term investments to total assets. To mitigate the impact of outliers, all ratios are winsorized at the 5th and 95th percentiles of their empirical distribution.

In addition to accounting variables, Table 3 also includes governance variables. *G-index* is the governance index of Gompers et al. (2003) in the calendar year that ends before the shareholder meeting. It adds one index point for each of 24 (anti)governance provisions. G-index is obtained from RiskMetrics. It is available for the years 1990, 1993, 1995, 1998, 2000, 2002, 2004, and 2006. To fill in the missing years, I use the latest available value of the index. Institutional owner*ship* is the percentage of shares owned by institutional investors in the quarter that ends prior to the date of the shareholder meeting. The data on institutional ownership are obtained from the Thomson-Reuters Institutional Holdings database. Finally, *inside owner*ship is the percentage of shares owned by all executives of the firm, which is obtained from ExecuComp in the fiscal year that ends prior to the shareholder meeting.

Table 3 also includes a measure of CSR: *KLD index*. This index is obtained from the Kinder, Lydenberg, and Domini (KLD) database. KLD is a social choice investment advisory firm that relies on independent rating experts to assess how well companies address the needs of their stakeholders based on multiple data sources including annual questionnaires sent to companies' investor relations offices, firms' financial statements, annual and quarterly reports, general press releases, government surveys, and academic publications. The composite *KLD index* is constructed by summing up the number of CSR strengths along the following dimensions: employees, customers, the natural environment, and society at large (community and minorities).⁶ KLD index is computed in the calendar year that ends prior to the shareholder meeting. As can be seen, the average *KLD index* is 4.06, which indicates that the average company in my sample has about four CSR strengths.

Finally, Table 3 also includes two industry-level measures of "stakeholder sensitivity," i.e., the extent

to which performance is sensitive to the relationship with employees and customers, respectively. Following Agrawal and Matsa (2013), I define labor *intensity* as the median wages-to-sales ratio in the company's two-digit Standard Industrial Classification (SIC) industry in the year preceding the vote. The wages-to-sales ratio is computed as the ratio of labor and pension expenses to sales using Compustat data. Since companies in labor-intensive industries rely more heavily on human capital, their performance is likely more sensitive to their relationship with employees. The second measure, B2C industry, is an indicator variable equal to 1 if the company operates in a business-to-consumer (B2C) industry, where individual consumers are the predominant customers. Lev et al. (2010) show that individual consumers are more sensitive to companies' social engagement than industrial buyers, which reflects inherent differences in the purchasing decision-making process (Corey 1991).⁷ To identify the set of B2C industries, I use the classification of Lev et al. (2010, p. 188) based on fourdigit SIC codes.

3.4. Tests for Quasi-Randomized Assignment

The identifying assumption of the RDD is that, around the majority threshold, passing a CSR proposal is as good as randomly assigned. This assumption has testable implications, akin to the tests of effective randomization in experimental data. In this section, I provide two standard tests of this assumption.

3.4.1. Continuity in the Distribution of Shareholder Votes. The first test evaluates whether the distribution of shareholder votes is continuous around the majority threshold. Any discontinuity would be symptomatic of a nonrandom assignment of "pass" versus "fail" around the threshold.

A visual inspection of the histogram in Figure 1 suggests that the distribution is indeed smooth and continuous around the majority threshold. A more formal approach is provided in Figure 2, which implements the McCrary (2008) test for smoothness of the density function around the threshold. As is shown, there is no evidence for a discontinuous jump. The null of continuity of the density function at the threshold cannot be rejected (*p*-value = 0.974).⁸

⁸ Finding a smooth distribution around the majority threshold is typical of shareholder-sponsored proposals (e.g., CGG, Cuñat

⁶ The KLD database also compiles strengths pertaining to corporate governance. Since the definition of CSR underlying this study does not encompass corporate governance, governance strengths are not included in the calculation of *KLD index*.

⁷ More precisely, "[t]he purchasing decision of an individual consumer is affected not only by product attributes, but also by social group forces, psychological factors, and the consumer's situational forces. In contrast, in industrial purchasing, the decision-making process is highly formalized, using defined procurement procedures, and subject to economic (cost/value) analysis" (Lev et al. 2010, p. 186; adapted from Corey 1991).

Table 4

Change from

Figure 2 Visualization of the McCrary (2008) Test



Notes. This figure presents a visualization of the McCrary (2008) test for the continuity of the vote share distribution around the majority threshold. The horizontal axis indicates the victory margin (i.e., the vote share minus the majority threshold). The vertical axis indicates the logarithm of the estimated density.

3.4.2. Preexisting Differences. The second testable implication of the randomness assumption is that companies whose voting share is immediately below or above the majority threshold should be very similar on the basis of ex ante characteristics. Intuitively, if the outcome of close call proposals is as good as randomized, it should be orthogonal to firm characteristics prior to the vote.

In Table 4, I examine whether there are any preexisting differences between companies that pass and reject CSR proposals. I consider all characteristics listed in Table 3. In columns (1) and (2), I examine these characteristics in the year preceding the shareholder meeting (t - 1). In columns (3) and (4), I examine the change in these characteristics between years t - 2 and t - 1. The exception is the abnormal return in the first row of the table, where t - 1 is the day prior to the meeting and the change from t - 2 to t-1 refers to the difference in abnormal returns in the two days preceding the meeting. Columns (1) and (3) report the differences among all firms in my sample, whereas columns (2) and (4) report the differences at the majority threshold by including third-order polynomials on both sides of the threshold.⁹

As can be seen in columns (1) and (3), companies that pass a CSR proposal differ significantly from companies that reject it. Importantly, however,

9 See CGG (pp. 1959-1961) for a similar test.

	Before meeting (t – 1)		•	e from o (t – 1)
	(1)	(2)	(3)	(4)
Abnormal return	-0.000	-0.002	0.001	0.003
	(0.006)	(0.006)	(0.008)	(0.009)
<i>Market value</i> (log)	-1.519*** (0.391)	-0.264 (0.642)	0.064 (0.075)	0.000 (0.161)
Total assets (log)	-1.579*** (0.346)	-0.375 (0.588)	0.003 (0.031)	0.009
ROA	-0.045**	-0.004	0.001	-0.008
	(0.020)	(0.030)	(0.006)	(0.010)
ROE	-0.088**	0.037	0.039	0.038
	(0.038)	(0.063)	(0.045)	(0.071)
NPM	-0.046	-0.026	0.005	-0.024
	(0.043)	(0.099)	(0.004)	(0.036)
Tobin's Q	-0.027	0.005	0.026	0.009
	(0.064)	(0.084)	(0.022)	(0.031
KLD index	-2.709***	-0.059	0.092	0.291
	(0.543)	(0.879)	(0.280)	(0.437
G-index	0.276	-1.057	-0.067	0.046
	(0.470)	(0.733)	(0.063)	(0.060
Institutional ownership (%)	8.388*	-0.817	-2.567**	-1.188
	(4.924)	(8.420)	(1.241)	(2.473
Inside ownership (%)	-0.355*	0.072	0.570	0.011
	(0.208)	(0.128)	(0.782)	(0.718
Labor productivity (log)	-0.015	-0.006	-0.003	-0.017
	(0.179)	(0.316)	(0.032)	(0.052
Capital expenditures	-0.009	0.004	-0.007**	-0.007
	(0.008)	(0.013)	(0.003)	(0.008
Sales growth	0.005	-0.004	0.027	-0.018
	(0.028)	(0.051)	(0.033)	(0.059
Leverage	-0.053	-0.093*	-0.001	0.020
	(0.033)	(0.053)	(0.016)	(0.028
Cash	0.051**	-0.004	-0.007	0.000
	(0.025)	(0.039)	(0.006)	(0.010
Labor intensity (industry-level)	0.021	0.000	0.023	0.006
	(0.037)	(0.049)	(0.016)	(0.022
B2C industry	0.044 (0.096)	0.006 (0.152)	_	_
Polynomial in vote share	No	Yes	No	Yes

Preexisting Differences as a Function of the Vote Outcome

Doforo montine

Notes. This table tests whether the adoption of CSR proposals is systematically related to firm characteristics prior to the meeting. All characteristics are defined in Table 3. In columns (1) and (2), these characteristics are measured in the year preceding the shareholder meeting (t - 1). Columns (3) and (4) consider the change in these characteristics between years t - 2 and t - 1. The exception is the abnormal return in the first row of the table, where t refers to days instead of years (e.g., t - 1 is the day prior to the meeting). Each row corresponds to a different characteristic and each entry comes from a separate regression. Columns (1) and (3) report the differences at the majority threshold by controlling for a third-order polynomial in the vote share on both sides of the threshold. Standard errors (in parentheses) are clustered at the firm level.

 $^{\ast}, \ ^{\ast\ast}, \ and \ ^{\ast\ast\ast}$ denote significance at the 10%, 5%, and 1% levels, respectively.

et al. 2013, Listokin 2008). This is in stark contrast to *management*sponsored proposals (which are excluded from my analysis). As Listokin (2008, p. 161) shows, for management-sponsored proposals, the distribution of votes exhibits a sharp discontinuity at the majority threshold. Essentially, these proposals rarely fail as management strategically withdraws those proposals that are expected to fail.

columns (2) and (4) show that these differences disappear at the majority threshold.¹⁰ Overall, this evidence suggests that there is no significant difference between companies that pass and reject CSR proposals around the majority threshold, which lends support to my identification strategy.

4. Results

4.1. Main Results

4.1.1. Graphical Analysis. To measure the impact of passing a CSR proposal on shareholder value, I examine the stock market reaction on the day of the shareholder meeting. Figure 3 provides a visualization of the data. Specifically, the figure plots abnormal returns against the victory margin (i.e., the vote share minus the majority threshold). Each dot in the figure represents the average abnormal return in 2% bins of vote share. The solid line plots predicted values of abnormal returns from third-order polynomials in vote share estimated separately to the left and right of the majority threshold.

As can be seen from the figure, abnormal returns appear to be a continuous and smooth function of the vote share everywhere except at the winning threshold, where there is a discontinuous jump. This evidence suggests that proposals that are approved by a small margin of votes lead to an increase in firm value compared with proposals that fail by a small margin of votes. Interestingly, abnormal returns seem to converge to zero as we move further to the left or right of the majority threshold. This pattern suggests that the outcome of non-close votes is anticipated by the market, and hence any value implication is already impounded in stock prices.¹¹

4.1.2. Regression Analysis. The graphical analysis in Figure 3 suggests that the adoption of close call CSR proposals leads to an increase in shareholder value. A more formal test of this hypothesis is provided in Table 5, which reports estimates of the difference in abnormal returns between CSR proposals that pass and CSR proposals that fail for increasingly small intervals around the majority threshold.

Column (1) of Table 5 estimates this difference in the full sample of 2,729 proposals and shows that it is insignificant. The lack of significance is driven by the non-close proposals (i.e., those proposals whose vote share is more than 10% above or below the majority threshold). As is shown in column (2), restricting the sample to non-close proposals yields a difference in abnormal returns that is virtually zero. This finding likely reflects the fact that the outcome of non-close proposals is highly predictable, and hence their effect is already incorporated in stock prices prior to the vote.

Column (3) restricts the sample to the 122 CSR proposals whose vote share is within 10% of the majority threshold. The difference in abnormal returns is 1.07%, which is significant at the 5% level. This difference is somewhat larger in columns (4)–(6), where the sample is restricted to proposals whose vote share is within 5%, 2.5%, and 1.5%, respectively, of the majority threshold (the difference in abnormal returns lies between 1.09% and 1.36%). It is significant in columns (4) and (5) and marginally insignificant (t = 1.58) in column (6). The weaker significance in column (6) is likely due to the small number of observations (only nine proposals have vote shares within 1.5% of the threshold). Overall, the evidence in columns (3)–(6) indicates that CSR proposals that are approved by a small margin of votes lead to a significant increase in shareholder value compared to CSR proposals that are marginally rejected.

In column (7), I estimate the specification given by Equation (1). This specification controls for two polynomials of order 3 in the vote share on both sides of the majority threshold. Unlike the nonparametric estimates in columns (3)–(6), this approach makes use of all 2,729 CSR proposals and hence provides a more efficient estimate of the causal effect of CSR proposals on abnormal returns. As is shown, the coefficient on the *pass* dummy is 1.18% and is significant at the 5% level. This coefficient is very similar to the estimate in column (6) that relies on proposals whose outcome is closest to the threshold.¹²

Finally, in column (8), I reestimate the regression in column (7) with control variables. The controls include all variables listed in Table 3 measured prior to the vote, as well as year dummies. If the outcome of the vote is truly random, including these controls should not affect the coefficient on the pass dummy as in randomized experiments—since all predetermined characteristics should be orthogonal to the assignment of pass versus fail. Indeed, I find that the coefficient is very similar to the one in column (7).

4.1.3. Multiple Votes and Multiperiod Analysis. As discussed in §3.2, the analysis in Table 5 does not account for the possibility that shareholders may have to vote on more than one CSR proposal in a given

¹⁰ Leverage in t - 1 is significantly different at the 10% level. However, given the large number of characteristics considered in Table 4, it is expected that some of them would appear significantly different even if the two groups of firms are drawn from the same distribution.

¹¹ This pattern is in line with the theoretical prediction of CGG (see their Figure 1 on p. 1951).

¹² The coefficient on the *pass* dummy is very similar if instead of third-order polynomials in the vote share, I use second- or fourth-order polynomials. The corresponding coefficients are 1.20% (t = 2.28) and 1.12% (t = 1.89), respectively.

Figure 3 Abnormal Returns on the Day of the Vote



Notes. The vertical axis indicates abnormal returns on the day of the vote. Abnormal returns are computed using the four-factor model of Carhart (1997). The horizontal axis indicates the victory margin (i.e., the vote share minus the majority threshold). Each dot in the figure represents the average abnormal return in 2% bins of victory margin. The solid line plots predicted values of abnormal returns from third-order polynomials in victory margin estimated separately to the left and right of the majority threshold.

meeting. Nor does it account for the possibility that the effect of the shareholder vote may affect returns beyond the day of the meeting.

To address these caveats, I estimate the specification given by Equation (2), i.e., the multiperiod specification in which multiple CSR proposals in a given meeting are "aggregated." This regression estimates the effect of passing a CSR proposal on abnormal returns on the meeting date (t), the day after (t + 1), and over the period from t + 2 to t + 7 (the cumulative abnormal returns over this period). The results are presented in column (1) of Table 6. As is shown, I only

Table 5 Abnormal Returns Around the Majority Threshold

find a significant effect on the day of the meeting (0.92% abnormal returns, $t = 2.56$), and the magnitude
of this effect is similar to the full model estimates in
Table 5. In the days following the meeting, the abnor-
mal returns are small and insignificant. This pattern
indicates that shareholders react on the day of the
vote, when the surprise around the threshold occurs.
In the remainder of this paper, I will refer to the spec-
ification used in column (1) of Table 6 as my base-
<i>line specification</i> since it is more conservative than the
specifications used in Table 5.

				Vote share				Full model
	All votes Non-close (1) (2)	±10% (3)	±5% (4)	±2.5% (5)	±1.5% (6)	Full model with co	with controls (8)	
Pass	0.0064	-0.0008	0.0107**	0.0136**	0.0109***	0.0117	0.0118**	0.0107**
	(0.0040)	(0.0044)	(0.0046)	(0.0058)	(0.0038)	(0.0074)	(0.0053)	(0.0052)
<i>R</i> -squared	0.001	0.000	0.056	0.099	0.255	0.204	0.007	0.044
Observations	2,729	2,607	122	61	23	9	2,729	1,780

Notes. This table presents regressions of the abnormal returns on the day of the vote on the *pass* dummy, i.e., a dummy variable that equals 1 if the proposal is adopted and 0 otherwise. Abnormal returns are computed using the four-factor model of Carhart (1997). In column (1), the sample consists of all 2,729 CSR proposals. Column (2) restricts the sample to non-close CSR proposals (i.e., those proposals whose vote share is more than 10% above or below the majority threshold). Columns (3)–(6) restrict the sample to CSR proposals whose vote share is within 10%, 5%, 2.5%, and 1.5%, respectively, of the majority threshold. Column (7) estimates the specification given by Equation (1) in the full sample (see §3.2). This specification controls for two polynomials of order 3 in the vote share on both sides of the majority threshold. Column (8) is a variant of the specification in column (7), except that the regression also includes control variables. The control variables include all variables listed in Table 3 measured prior to the vote, as well as year dummies. Standard errors (in parentheses) are clustered at the firm level.

*, **, and *** denote significance at the 10%, 5%, and 1% levels, respectively.

				Confounding effect	of governance proposals?		
		Robu	stness	No governance	No governance	Companies wit	th low vs. high CSR
	(1)	Market model (2)	Positive returns (3)	proposal $\pm 10\%$ (4)	proposal ±20% (5)	Low KLD index (6)	High KLD index (7)
Day of vote (t)	0.0092** (0.0036)	0.0093** (0.0037)	0.2685*** (0.0871)	0.0101** (0.0041)	0.0109*** (0.0046)	0.0102** (0.0047)	0.0054* (0.0031)
One day later $(t + 1)$	-0.0009 (0.0023)	-0.0004 (0.0025)	0.0973 (0.0635)	0.0005 (0.0027)	0.0004 (0.0027)	0.0011 (0.0031)	-0.0039 (0.0029)
Days $t + 2$ to $t + 7$	-0.0043 (0.0092)	-0.0019 (0.0099)	0.0540 (0.0741)	-0.0050 (0.0102)	-0.0044 (0.0104)	-0.0061 (0.012)	-0.0004 (0.0092)
<i>R</i> -squared Observations Number of firm-meetings	0.043 9,225 1,845	0.042 9,225 1,845	0.016 9,225 1,845	0.009 7,615 1,523	0.007 6,640 1,328	0.021 4,215 843	0.088 4,220 844

Table 6 Abnormal Returns of Passing CSR Proposals

Notes. The regression in column (1) estimates the effect of passing a CSR proposal on the four-factor abnormal returns on the day of the vote (t), the following day (t + 1), and over the period from t + 2 to t + 7 (the cumulative abnormal returns over this period). Columns (2)-(7) present variants of the regression in column (1). In column (2), abnormal returns are computed using the market model instead of the four-factor model. In column (3), the dependent variable is a dummy variable that equals 1 if the (four-factor) abnormal return is positive and 0 otherwise. In columns (4) and (5), the sample excludes all shareholder meetings in which a governance proposal received a vote share within 10% and 20%, respectively, of the majority threshold. In columns (6) and (7), the sample is restricted to companies whose KLD index in the year preceding the vote lies below and above, respectively, the median across all firms in the same two-digit SIC industry. The regression specification used in all columns is the multiperiod specification given by Equation (2), in which multiple CSR proposals in a given meeting are aggregated (see §3.2). All regressions include firm-meeting fixed effects, distance-to-the-election fixed effects, as well as year fixed effects. Standard errors (in parentheses) are clustered at the firm level.

*, **, and *** denote significance at the 10%, 5%, and 1% levels, respectively.

4.1.4. Robustness. Columns (2) and (3) of Table 6 present two robustness checks. These robustness checks are variants of the baseline specification used in column (1).

In column (2), I compute abnormal returns using the market model instead of the four-factor model. As can be seen, all coefficients are very similar to before: the effect on the day of the vote is 0.93%, whereas there is no significant effect in the following days.

Given the small number of close call CSR proposals, a potential concern is that my results may be driven by a few large abnormal returns around the majority threshold. To address this concern, I reestimate my baseline specification replacing the abnormal returns by a dummy variable that equals 1 if the abnormal return is positive and 0 otherwise. Since this dummy ignores the magnitude of the abnormal return, it is not sensitive to outliers. (See CGG for a similar robustness check.) As is shown in column (3), the pattern is similar to before. In particular, on the day of the vote, CSR proposals that pass by a small margin of votes are 27% more likely to have positive abnormal returns compared with CSR proposals that fail marginally.

4.1.5. Confounding Effect of Governance Proposals? Another potential concern is that shareholders may not only vote on CSR proposals but also on governance proposals during the same meeting. As CGG show, governance proposals are very frequent, and the adoption of close call governance

proposals generates substantial abnormal returns. If, for some reason, shareholders tend to vote in a similar way on all proposals, it could be that CSR proposals that pass by a small margin of votes tend to occur in meetings in which governance proposals also pass by a small margin. In this case, my results may be capturing some of the effect of governance proposals. To address this concern, I reestimate my baseline specification after excluding all shareholder meetings in which a governance proposal received a vote share within 10% of the majority threshold. Arguably, a governance vote that easily passes or is easily defeated has little uncertainty surrounding it, and hence it should not affect the stock price reaction. By contrast, close call governance proposals are likely to affect stock prices. As is shown in column (4) of Table 6, my results are robust to this exclusion. In column (5), I further verify that my results hold if I exclude shareholder meetings in which a governance proposal received a vote within 20% of the majority threshold.

4.1.6. Level of CSR. The results presented so far suggest that the passing of close call CSR proposals improves CFP, which is consistent with the view that CSR is a valuable resource for firms. That being said, the "stock" of CSR resources that is already in place may influence the benefits from implementing an additional CSR program; i.e., the CSR–CFP relationship may not be linear. Arguments can be made for either a concave or convex relationship.

12

An argument in the spirit of neoclassical economic theory can motivate a concave relationship. Neoclassical models of the firm typically assume decreasing marginal returns of the production factors (e.g., capital and labor). By the same reasoning, CSR as a resource may exhibit decreasing marginal returns. Intuitively, in early stages of designing CSR policies, it may be fairly easy and inexpensive for companies to implement social programs that yield substantial monetary benefits. However, once the low-hanging fruits of CSR have been harvested, it may become increasingly difficult to adopt social policies that further improve the company's financial performance (for a related argument in the context of green initiatives, see Flammer 2013).

On the other hand, it could also be that CSR only affects CFP once a "critical mass" of CSR resources is in place, in which case the relationship is convex. Edmans (2011, 2012) provides evidence along these lines: he shows that there are benefits to being in the very top level of employee satisfaction (the top 100, rather than merely above average).

Accordingly, establishing the precise shape of the CSR–CFP relationship is an empirical question. The empirical framework used in this paper is helpful in addressing this question. Specifically, in columns (6) and (7) of Table 6, I reestimate my baseline specification separately for companies whose KLD index is below or above the median across all firms in the same two-digit SIC industry (in the year that ends prior to the vote). As is shown, the effect is about twice as strong for companies with a lower KLD index. This implies that companies with stronger social performance benefit less from the passing of an additional CSR resolution, thus pointing at a concave relationship between CSR and CFP.

4.2. Magnitude of the Effect

In my baseline specification, the abnormal return of passing a close call CSR proposal is 0.92%. Since shareholder proposals are not binding, this estimate only measures the effect of approving a CSR proposal as opposed to the effect of *implementing* a CSR proposal. The latter can be approximated by rescaling the estimated coefficient by the probability of implementing the proposal. In §4.3.1, I show that CSR proposals have a 52% probability of being implemented. Accordingly, the approximate effect of implementing a close call CSR proposal is 0.92/0.52 = 1.77% in abnormal returns.¹³

It is helpful to compare this effect with the effect of other shareholder proposals that have been studied in the literature. In their study of governance proposals, CGG find that adopting close call governance proposals leads to an increase in shareholder value by 2.8%. In a related study, Cuñat et al. (2013) show that the adoption of close call "say-on-pay" proposals increases shareholder value by 4.6%. This comparison indicates that, although CSR proposals are beneficial to shareholders, the benefits are not as substantial as those from governance-type proposals.

Although the value gains from CSR proposals may seem small compared with other proposals, they are nevertheless economically significant. This can be seen by benchmarking my results against the findings from other CSR studies. In particular, Klassen and McLaughlin (1996) find abnormal returns of 0.82% around the announcement of environmental awards. Similarly, Flammer (2013) finds abnormal returns of 0.84% around the announcement of green initiatives. Finally, Edmans (2012) shows that companies listed in the "100 Best Companies to Work For in America" generated 2.3% to 3.8% higher abnormal returns per year from 1984 through 2011. Overall, my estimate of 1.77% lies within the ballpark of what these studies have found.

4.3. Long-Run Effects of CSR

In this section, I evaluate the effect of adopting a close call CSR proposal on long-term firm outcomes (as opposed to the short-term stock market reaction). The specification used throughout is the specification given by Equation (2) at annual frequency. This specification estimates the effect of passing a close call CSR proposal on a given outcome variable in the year of the proposal (*t*), the following year (*t* + 1), and the subsequent three years (the average of the outcome variable from t + 2 to t + 4). All results are presented in Table 7.

4.3.1. Implementation. As mentioned in §4.2, shareholder proposals are not binding. Although data on whether each proposal was implemented are not publicly available, I do observe whether the KLD index (i.e., the number of CSR strengths) has

¹³CGG propose an alternative way of computing the (implementation-adjusted) effect of a close call proposal on abnormal returns that takes into account not only the probability of the proposal being implemented (p^{l}) but also the additional effect on the submission and implementation of future proposals.

Specifically, instead of computing the ratio β/p^l , one would compute $\beta/(p^l + \sum_{i=1}^{\infty} \delta^i p_{l+i}^p)$, where δ is the discount rate and p_{l+i}^p is the change in the probability of passing and subsequently implementing another CSR proposal *i* periods from now. Using a discount rate of 5% and the forward-looking implementation probabilities provided in column (1) of Table 7, I find that the corresponding abnormal return is 1.76%, which is almost identical to the 1.77% estimate provided above (see pp. 1969–1970 of CGG for a similar calculation).

	Implementation KLD index (1)	Governance	Performance				Mechanism						
						KLD index passed	Proposal passed (2)	<i>ROA</i> (3)	NPM (4)	<i>ROE</i> (5)	Tobin's Q (6)	Sales growth (7)	Labor productivity (8)
Year of the meeting (t)	0.521**	0.0326	0.0018	0.0017	0.0052	0.0195**	0.0089	0.0141	-0.0002				
	(0.213)	(0.0238)	(0.0019)	(0.0026)	(0.0069)	(0.0079)	(0.0083)	(0.0176)	(0.0025)				
One year later $(t + 1)$	0.541**	-0.0023	0.0043*	0.0046*	0.0091	0.0215**	0.0155*	0.0437***	0.0008				
	(0.220)	(0.0187)	(0.0023)	(0.0026)	(0.0077)	(0.0085)	(0.0089)	(0.0156)	(0.0026)				
Years $t + 2$ to $t + 4$	0.429**	-0.0007	0.0050**	0.0052*	0.0115	0.0158**	0.0132*	0.0373*	0.0001				
	(0.207)	(0.0541)	(0.0024)	(0.0030)	(0.0082)	(0.0078)	(0.0080)	(0.0194)	(0.0024)				
<i>R-</i> squared	0.413	0.429	0.845	0.915	0.559	0.859	0.360	0.948	0.849				
Observations	7,653	9,225	8,291	8,291	7,322	7,283	8,388	8,364	8,266				
Number of firm-meetings	1,689	1,845	1,815	1,815	1,651	1,675	1,803	1,837	1,819				

Table 7 Long-Run Effects of CSR Proposals

Notes. The regressions presented in this table estimate the effect of passing a CSR proposal on several firm outcomes in the year of the vote (t), the following year (t + 1), and the subsequent three years (average of the outcome variable from t + 2 to t + 4). The dependent variables in columns (1)-(9) are *KLD index*, a dummy variable equal to 1 if a shareholder proposal on governance is passed (*proposal passed*), *ROA*, *NPM*, *ROE*, *Tobin's Q*, *sales growth*, *labor productivity* (in logarithm), and *capital expenditures*, respectively. All variables are defined in Table 3. The regression specification used in all columns is the multiperiod specification given by Equation (2). All regressions include firm-meeting fixed effects, distance-to-the-election fixed effects, as well as year fixed effects. Standard errors (in parentheses) are clustered at the firm level.

*, **, and *** denote significance at the 10%, 5%, and 1% levels, respectively.

increased after the shareholder meeting. Thus, studying the changes in the KLD index provides a rough estimate of the implementation probability.¹⁴

The results with the KLD index as dependent variable are presented in column (1). As is shown, the KLD index increases by 0.52 CSR strengths in the year of the proposal (i.e., the year that ends after the meeting date), which can be interpreted as a 52% implementation likelihood. The KLD index remains around that level in the subsequent years.¹⁵

4.3.2. Corporate Governance. An alternative interpretation of my findings is that the passing of CSR proposals may encourage the adoption of subsequent governance proposals, and it is these proposals that lead to improvements in financial performance. To address this alternative interpretation, I examine whether the passage of CSR proposals affects the likelihood of adopting governance proposals. Specifically, I use as the dependent variable an indicator variable equal to 1 if a governance resolution is adopted. As is shown in column (2), the likelihood of passing a governance proposal does not increase significantly in

the four years following the passage of close call CSR proposals.¹⁶

4.3.3. Operating Performance and Firm Value. Next, I examine the effect of passing a close call CSR proposal on operating performance. I consider three measures of operating performance: *ROA*, *NPM*, and *ROE*.

The results for *ROA* are provided in column (3). As can be seen, ROA increases in the year of the meeting. However, the increase is small and insignificant. In the first year after the meeting, the increase in *ROA* is larger and statistically significant. It is also economically significant: the reported coefficient of 0.0043 corresponds to an increase in ROA by 3.1%, which represents 5% of the standard deviation of ROA (mean and standard deviation of ROA are 0.137 and 0.083, respectively; see Table 3). In years t + 2 to t + 4, the effect remains stable. Overall, the dynamic pattern indicates that it takes 12-24 months for the CSR program to materialize into higher profitability, and this effect appears to be long-lasting. The pattern is qualitatively similar when I look at NPM and ROE in columns (4) and (5), although the coefficients in the *ROE* regression are not significant.

Finding a long-lasting improvement in operating performance indicates that the increase in shareholder value—which represents the sum of all discounted future cash flows—may be permanent as well. To examine whether this is the case, I use *Tobin's Q* as

¹⁴ See CGG (p. 1968) for a similar approach. In their analysis of governance proposals, CGG look at changes in the *G*-index of Gompers et al. (2003)—the number of antitakeover provisions in place—to evaluate whether a governance proposal was implemented or not. ¹⁵ The 52% implementation likelihood is in the ballpark of what prior studies have found for other types of shareholder proposals. CGG and Ertimur et al. (2010) find that governance proposals that pass have an implementation likelihood of 31.3% and 31.1%, respectively. Cuñat et al. (2013) find that say-on-pay proposals that are approved have a 52.5% probability of being implemented.

¹⁶ Ideally, I would use the *G*-index as the dependent variable since it reflects actual changes in governance. However, the latest available year of the *G*-index is 2006, and most close call CSR proposals occur as of that year (see panel A of Table 2).

the dependent variable in column (6). As is shown, *Tobin's Q* increases already in the year of the meeting (which is likely driven by the stock market reaction around the day of the vote). In the subsequent years, it remains at a similar level. This pattern indicates that the value gains from close call CSR proposals are long-lasting.

4.3.4. How Does CSR Benefit Companies? The evidence reported so far suggests that CSR benefits shareholder value through an increase in operating performance. In the following, I discuss several plausible channels through which CSR may improve operating performance.

First, implementing CSR programs may be a way to cater to customers that are responsive to sustainable practices (e.g., Baron 2008, McWilliams and Siegel 2001, Reinhardt 1998). In this case, one might expect an increase in sales growth following the adoption of CSR proposals, as they would allow companies to extend their customer base.

Second, it could be that CSR programs increase employee satisfaction. Several articles in the management literature argue that by nurturing the relationship to its employees, a company can attract, motivate, and retain the most talented employees in the industry (e.g., Albinger and Freeman 2000, Greening and Turban 2000, Peterson 2004, Pfeffer 1994, Turban and Greening 1996, Vogel 2005). Relatedly, Edmans (2011, 2012) shows that companies with higher job satisfaction earn higher abnormal returns compared with their peers. Accordingly, if employee satisfaction is the underlying mechanism behind the increase in profitability, I should observe an increase in labor productivity following the passage of close call CSR proposals.

Third, CSR initiatives may foster the use of more efficient technologies or production processes (e.g., environment-friendly technologies). In particular, Porter (1991) views pollution as a waste of resources (e.g., energy and material) and argues that efforts to reduce pollution might not only reduce a company's environmental footprint but also strengthen its competitiveness. A growing literature extends Porter's view (for reviews, see Ambec and Lanoie 2008, Berchicci and King 2007, Etzion 2007). For instance, the literature on sustainability in business examines ways in which companies can become more environmentally friendly and how these greening initiatives influence financial performance. In particular, companies can become more sustainable by leveraging the low-hanging fruits of efficiency and waste management and hence achieve significant financial benefits (e.g., Clelland et al. 2000, Rusinko 2007, Russo and Harrison 2005). Accordingly, if the increase in performance comes from improved production processes and technologies, I should observe an increase in capital investment after the passing of close call CSR proposals.

I examine these three channels in Table 7, columns (7)–(9). As is shown in column (7), sales growth increases after the adoption of close call CSR proposals, which is consistent with the demand channel. As for the supply-side mechanisms, the estimates in column (8) show that labor productivity increases significantly in the years following the vote, which lends support to the employee satisfaction channel. By contrast, the estimates in column (9) show no evidence for a significant increase in physical investment. The latter finding does not necessarily reject the investment channel. For example, it could be that companies' efforts to reduce emissions translate in a decrease in investment (e.g., the closure of polluting facilities) so that the net effect is ambiguous. A more thorough analysis of this mechanism would require detailed facility-level data.¹⁷

5. External Validity

5.1. How Representative Is the Sample?

Are the companies in my sample representative of the average public firm in the U.S. economy? To assess whether this is the case, I benchmark my sample with the Compustat universe. The comparison is provided in panel A of Table 8. The column under the heading "Mean CSR proposals" restates the means from Table 3, and the column under the heading "Mean Compustat" provides the corresponding means based on all firm-year observations in Compustat during the sample period (1997–2012). The column next to it reports the *p*-value of the difference-in-means test.

As can be seen, the two samples differ on the basis of several characteristics. Two main differences are worth emphasizing. First, companies in my sample are significantly larger.¹⁸ This pattern is similar to the Cuñat et al. (2013) finding that companies whose shareholders submit say-on-pay proposals are on average larger than S&P 1500 companies. Second, companies in my sample display higher levels of CSR: their number of KLD strengths is about four

¹⁷ In untabulated regressions, I also examine whether companies increase their marketing expenses after the passage of close call CSR proposals, e.g., as a way to communicate their CSR efforts to their customer base. Specifically, I consider advertising expenses (defined as the ratio of advertising expenses to total assets from Compustat) as the dependent variable. I find no significant change in this variable after the vote.

¹⁸ To the extent that the main sampling frame of SharkRepellent and RiskMetrics are the Russell 3000 and the S&P 1500 indices, respectively, this difference is partly mechanical. However, it does subsist if instead of the Compustat universe I use the Russell 3000 universe as the benchmark.

Table 8 External Validity

	CSR proposals	Compustat vs. CSR	proposals	Close call CSR proposa	als vs. CSR proposals
	Mean CSR proposals	Mean Compustat	<i>p</i> -value	Mean close call	<i>p</i> -value
	F	Panel A: Firm characteristic	S		
Market value (\$ million)	37,881	3,266	0.000	28,639	0.126
Total assets (\$ million)	77,365	8,868	0.000	63,686	0.267
ROA	0.137	0.049	0.000	0.128	0.147
ROE	0.362	0.144	0.000	0.346	0.596
NPM	0.194	0.045	0.000	0.182	0.522
Tobin's Q	1.890	1.905	0.904	1.801	0.363
KLD index	4.06	1.26	0.000	3.65	0.373
G-index	9.23	8.99	0.159	9.91	0.294
Institutional ownership (%)	63.27	36.49	0.000	75.61	0.016
Inside ownership (%)	0.64	1.39	0.000	0.81	0.589
Labor productivity	437	323	0.000	526	0.424
Capital expenditures	0.051	0.133	0.000	0.050	0.826
Sales growth	0.071	0.022	0.000	0.091	0.441
Leverage	0.264	0.214	0.000	0.246	0.757
Cash	0.101	0.181	0.000	0.109	0.697
Labor intensity (industry level)	0.321	0.258	0.000	0.399	0.036
B2C industry	0.425	0.347	0.000	0.541	0.047
	Pai	nel B: Proposal characteris	tics		
Labor and environment issues	0.404	·	_	0.656	0.000
Related to performance	0.224	_	_	0.557	0.000

Notes. This table compares the sample of CSR proposals ("CSR proposals") with the Compustat universe ("Compustat") and the subset of CSR proposals whose vote outcome is within 5% of the majority threshold ("Close call CSR proposals"). Means for the CSR proposal sample are restated from Table 3. Means for the Compustat sample are computed across all available firm-year observations during the sample period (1997–2012). Means for the close call CSR proposals sample are computed as in Table 3, but with respect to the close call CSR proposals. The *p*-values refer to the difference-in-means test. The variables in panel A are defined in Table 3. *Labor and environment issues* is an indicator variable equal to 1 if the proposal is on labor or environment issues (see panel B of Table 2). *Related to performance* is an indicator variable equal to 1 if the proposal's support statement contains performance keywords (see §5.2).

times higher. This difference is intuitive: companies whose shareholders are more inclined toward CSR have higher levels of CSR and at the same time are more likely to submit CSR proposals. Although these differences do not bias my estimate of the treatment effect, they do indicate that my findings may not necessarily apply to the average U.S. public firm.

5.2. How Representative Are Close Call CSR Proposals?

A limitation of my research design is that the effect is identified by the subset of CSR proposals whose vote outcome is close to the majority threshold. Although this limitation is inherent to any RDD, it is especially important in my setting given the relatively small number of close call CSR proposals. As discussed in §3.1, only 61 out of the 2,729 proposals have vote outcomes within 5% of the majority threshold; i.e., about 2.2% of the proposals identify the effect at work.¹⁹

Moreover, the vast majority of CSR proposals receive little support at shareholder meetings (the average vote outcome is merely 13.5%; see §3.1). This is in sharp contrast to close call proposals, which garner sufficient support to achieve a vote outcome close to 50%. Hence, close call CSR proposals appear to be "special" and hence may not be representative of the average CSR proposal. In the following, I characterize the set of close call CSR proposals and discuss how they differ from non-close proposals.

First, I examine which types of proposals (based on the nine CSR subcategories listed in panel B of Table 2) are more frequently found among the 61 close call proposals. The subcategory that is most frequently represented is labor issues (23 proposals). The two subcategories pertaining to environment issues (environmental issues and sustainability report) account for 17 proposals. The remaining 21 proposals are scattered among the other six subcategories (add minorities/women to board, animal rights, health

¹⁹ As discussed in §4.2, shareholder proposals are typically not binding; i.e., even if a proposal is formally approved, it may not be implemented. Accordingly, another potential caveat is that only those proposals that are eventually implemented (the "compliers," using the terminology of Imbens and Lemieux 2008) may contribute to the identification. This caveat is mitigated in my main analysis, since I focus on the stock market reaction on the day of the

vote. On that day, shareholders do not know with certainty whether an approved proposal will be implemented or not. Accordingly, as long as the implementation probability is strictly positive, even the (eventual) "noncompliers" contribute to the stock market reaction (see Section I.B of CGG for a similar argument).

issues, human rights, other social issues related, and political issues). Accordingly, about two-thirds (65.6%) of the close call proposals concentrate among three subcategories pertaining to labor and environment issues. To further zoom into these three subcategories, I review the proposal description in the proxy statements (SEC Form DEF 14A). Virtually all proposals on labor issues are employee satisfaction initiatives in the form of nondiscrimination and fairness policies. As for the proposals on environment issues, most of them pertain to the mitigation of environmental hazards.

That 65.6% of the close call CSR proposals cluster among labor and environment issues is not a typical feature of CSR proposals. Across all CSR proposals, the share of labor and environment proposals is only 40.4%. As can be seen from panel B of Table 8, this difference is highly significant (*p*-value = 0.000).

Next, I examine whether close call CSR proposals are more likely to be perceived as value-enhancing by shareholders. To address this question, I conduct a textual analysis of the support statement provided in the proxy statement of the proposals.20 A caveat of this analysis is that it is inherently subjective, yet it helps shed some light on shareholders' motives behind these proposals. Specifically, I search for keywords related to performance. These keywords include "profits," "performance," "productivity," "value," and "competitive" as well as variations thereof (e.g., for the word "profits," I also search for "profitability," "profitable," etc.). To ensure that each keyword is used in the (financial) performance context, I review each match manually. As can be seen in panel B of Table 8, I find that 55.7% of the close call proposals contain performance keywords. (This percentage is very similar on either side of the majority threshold.) In contrast, only 22.4% of all CSR proposals include such keywords. The difference is highly significant (p-value = 0.000). This analysis indicates that close call CSR proposals are more likely to have performance implications than are non-close proposals.²¹

²⁰ For example, in the support statement of a proposal to implement a nondiscrimination policy, shareholders of Gardner Denver, Inc., argue as follows: "Employment discrimination on the basis of sexual orientation and gender identity diminishes employee morale and productivity. Because state and local laws are inconsistent with respect to employment discrimination, our company would benefit from a consistent, corporate-wide policy to enhance efforts to prevent discrimination, resolve complaints internally, access employees from the broadest talent pool, and ensure a respectful and supportive atmosphere for all employees. Gardner Denver will enhance its competitive edge by joining the growing ranks of companies guaranteeing equal opportunity for all employees" (Gardner Denver 2010, p. 27). The expressions "productivity" and "competitive edge" point at the performance implications of this proposal.

²¹ Performance keywords are found more frequently in proposals on labor and environment issues, which may explain why close call To further characterize the set of close call CSR proposals, I examine whether companies around the discontinuity differ systematically from companies far from the discontinuity. This comparison is provided in panel A of Table 8. As can be seen, companies close to the threshold are significantly more likely to operate in labor-intensive and B2C industries (*p*-values of 0.036 and 0.047, respectively)—that is, in industries where performance is likely more sensitive to the relationship with employees and customers, respectively. This echoes the previous finding that close call CSR proposals are tied more closely to financial performance than are non-close proposals.²²

In sum, the analysis presented in this section suggests that close call CSR proposals are not representative of the average CSR proposal. Accordingly, one must be careful with extrapolation—my results do not necessarily generalize to CSR proposals far from the discontinuity. In particular, the link to performance, which appears to be more prevalent among close call CSR proposals, may explain why these proposals (1) receive larger shareholder support and (2) lead to positive announcement returns and superior accounting performance when adopted.

6. Conclusion

In this paper, I present evidence on the causal effect of CSR on financial performance. To obtain exogenous variation in CSR, I exploit the passage of shareholder proposals on CSR that pass or fail by a small margin of votes. The outcome of such close call proposals is as good as random and hence provides a randomized assignment of CSR to companies. Using an RDD methodology, I find that the adoption of close call CSR proposals leads to a significant increase in shareholder value by 1.77%. This finding is consistent with the view that CSR is a valuable resource, which is in line with, e.g., the resource-based view of the firm (e.g., Hart 1995, Russo and Fouts 1997),

CSR proposals cluster among these categories. Specifically, performance keywords appear in 59% of all labor proposals, 33% of all environment proposals, and 8% of all other proposals; for close call proposals, these ratios increase to 91%, 65%, and 9%, respectively. Given this pattern, one may expect my results to be stronger for labor and environment proposals. When I reestimate my baseline specification separately for the three different groups, I find that the stock market reaction is indeed larger for proposals on labor and environment issues. Given the relatively small number of close call proposals in each group, such an analysis is only suggestive, though.

²² There is no significant difference along the other characteristics considered in Table 8, with the exception of institutional ownership, which is higher for companies close to the threshold. This difference is related to the Karpoff et al. (1996) finding that institutional ownership affects the submission of shareholder proposals.

instrumental stakeholder theory (e.g., Jones 1995), and Porter and Kramer's (2006, 2011) shared value argument. I also find that the value gains are stronger for firms with relatively low levels of CSR prior to the vote. This suggests that CSR is a resource with decreasing marginal returns; i.e., the CSR–CFP relationship is concave.

I then examine the mechanisms through which CSR increases shareholder value. I find that the passing of close call CSR proposals has a positive impact on operating performance (return on assets, net profit margin, and return on equity). When I further examine what explains the increase in operating performance, I find that the adoption of close call CSR proposals has a positive impact on labor productivity and sales growth. This evidence suggests that these proposals improve employee satisfaction and help companies cater to customers that are sensitive to sustainable practices.

To the best of my knowledge, this study is the first to provide empirical evidence on the causal effect of CSR on CFP. Although a large empirical literature points toward a positive correlation between CSR and CFP (for detailed reviews of this literature, see Margolis and Walsh 2001, 2003; Margolis et al. 2007; Orlitzky et al. 2003; Roman et al. 1999), this correlation does not warrant a causal interpretation. The papers that are closest to making causal statements are Dowell et al. (2000), Lev et al. (2010), Preston and O'Bannon (1997), Waddock and Graves (1997), and Edmans (2011, 2012). The first four articles rely on so-called Granger causality tests (i.e., the comparison of leads and lags of CSR and CFP). However, as Lev et al. emphasize, although this methodology has its merits, it does not establish causality.²³ Accordingly, it does not substitute for the need to look for (quasi-)natural experiments such as the one I consider in this study. Edmans (2011, 2012) uses a different approach. He constructs a value-weighted portfolio of the "100 Best Companies to Work For in America" and shows that this portfolio significantly outperforms industry benchmarks. An appealing feature of his approach is that the effect of observable variables that are correlated with employee satisfaction is likely to be already impounded into the stock prices, which helps address causality. That being said, employee satisfaction may correlate with unobservable firm characteristics that investors have not fully incorporated into stock prices.

Finally, a limitation of my research design is that the identifying variation comes from a relatively small number of proposals—the ones close to the threshold—and those proposals may not be representative of the average CSR proposal. In particular, the large majority of CSR proposals receive little support at annual meetings, suggesting that shareholders see little value in them. By contrast, the subset of CSR proposals that make it close to the 50% threshold receive an unusually high support. This indicates that close call CSR proposals are somewhat special.

When I characterize close call CSR proposals, I find that—compared with proposals away from the threshold-they are more likely to address employee satisfaction and the mitigation of environmental hazards. Also, a textual analysis of their support statement shows that they more frequently contain performance-related arguments. Finally, close call proposals are more frequently found among companies operating in "stakeholder-sensitive" industries, i.e., industries in which performance depends greatly on the relationship with employees and customers. These differences indicate that close call CSR proposals are more closely tied to performance, which may explain why they are valueenhancing. Accordingly, one has to be careful in extrapolating my results-although the value gains documented in this paper apply to close call CSR proposals, they may not apply to CSR proposals in general. Extending the external validity of this study by identifying natural experiments that apply to a broader universe of firms and CSR initiatives is an exciting and challenging avenue for future research.

Supplemental Material

Supplemental material to this paper is available at http://dx .doi.org/10.1287/mnsc.2014.2038.

Acknowledgments

The author thanks Wei Jiang (the department editor), an anonymous associate editor, two anonymous reviewers, Alex Edmans, Xavier Giroud, Maria Guadalupe, Andrew Hoffman, Aleksandra Kacperczyk, Lloyd Kurtz, James Naughton, Felix Oberholzer-Gee, and Brian Richter; as well as seminar participants at Ivey Business School, the Alliance for Research on Corporate Sustainability Conference at the University of California at Berkeley (2013), and the SRI Conference on Sustainable, Responsible, Impact Investing (Colorado Springs, Colorado) for helpful comments and suggestions. The author also thanks Hannarae Annie Nam for excellent research assistance. Financial support from MIT's Undergraduate Research Opportunities Program is gratefully acknowledged.

²³ As Leamer (1985) points out, Granger causality does not imply "causality" but rather "precedence" in a lead–lag relationship. In the CSR–CFP context, finding that CSR precedes (i.e., predicts future values of) CFP does not necessarily imply that CSR causes CFP. For example, it could be that companies engage in CSR because they expect their future profitability to be higher. In this scenario, CSR predicts future CFP, yet there is no causal relationship between the two.

Appendix. Types of CSR Proposals

Proposal type	Proposal description
	Panel A: Environment issues
Environmental issues	Shareholder-sponsored proposals to request that the board issue a report detailing the company's impact on the environment or to request that the board adopt policies to minimize the company's negative impact on the environment. If a proposal combines health and environmental issues, SharkRepellent will generally classify it in the "health issues" category. If a proposal focuses on preparing a sustainability report regarding environmental practices, SharkRepellent will generally classify it in the "sustainability report" category.
Sustainability report	Shareholder-sponsored proposals to request that the board issue a report describing the company's strategies to ensure sustainability, usually focusing on actions to address greenhouse gas emissions and other environmental and social considerations.
	Panel B: Social issues
Add minorities/women to board	Shareholder-sponsored proposals to request that the board take steps to ensure that women and minority candidates are in the pool from which board nominees are chosen.
Animal rights	Shareholder-sponsored proposals to encourage the company to consider animal interests throughout its production and business processes or to request that the board adopt an animal welfare policy. People for the Ethical Treatment of Animals has submitted the majority of these proposals.
Health issues	Shareholder sponsored proposals to request that the board institute policies to protect human health or that the board issue a report regarding the company's stance on certain health-related issues.
Human rights	Shareholder-sponsored proposals to request that the board institute policies to protect and/or promote human rights. Such actions could include respecting human rights throughout the company's production process or refusing to do business with countries or businesses that contribute to human rights abuses.
Labor issues	Shareholder-sponsored proposal to request that the board institute certain labor-related policies. Such labor policies may include prohibiting discrimination based on sexual orientation and gender identity or abiding by certain fairness principles.
Other social issues related	Shareholder-sponsored proposal to request that the board provide a report regarding certain social issues. Common topics may include the examination of the company's effect on national security, the safety of the company's operations from terrorists attacks, and the company's lending practices.
Political issues	Shareholder-sponsored proposals to request that the board provide a report detailing the company's policies regarding political contributions.

Note. This table provides a description of the nine CSR subcategories from the SharkRepellent database (source: SharkRepellent).

References

- Agle BR, Donaldson T, Freeman RE, Jensen MC, Mitchell RK, Wood DJ (2008) Dialogue: Towards superior stakeholder theory. *Bus. Ethics Quart.* 18(2):153–190.
- Agrawal AK, Matsa DA (2013) Labor unemployment risk and corporate financing decisions. J. Financial Econom. 108(2):449–470.
- Albinger HS, Freeman SJ (2000) Corporate social performance and attractiveness as an employer to different job seeking populations. J. Bus. Ethics 28(3):243–253.
- Ambec S, Lanoie P (2008) Does it pay to be green? A systematic overview. Acad. Management Perspect. 22(4):45–62.
- Barney J (1991) Firm resources and sustained competitive advantage. J. Management 17(1):99–120.
- Baron DP (2008) Managerial contracting and corporate social responsibility. J. Public Econom. 92(1–2):268–288.
- Berchicci L, King AA (2007) Postcards from the edge: A review of the business and environment literature. *Acad. Management Ann.* 1(1):513–547.
- Brokaw L (2012) Marks and Spencer's emerging business case for sustainability. *MIT Sloan Management Review* (blog) (July 13),

http://sloanreview.mit.edu/article/marks-and-spencers-emerging -business-case-for-sustainability/.

- Carhart MM (1997) On persistence in mutual fund performance. J. Finance 52(1):57–82.
- Clelland IJ, Dean TJ, Douglas TJ (2000) Stepping towards sustainable business: An evaluation of waste minimization practices in US manufacturing. *Interfaces* 30(3):107–124.
- Corey ER (1991) Industrial Marketing Cases and Concepts (Prentice Hall, Englewood Cliffs, NJ).
- Cuñat V, Giné M, Guadalupe M (2012) The vote is cast: The effect of corporate governance on shareholder value. J. Finance 67(5):1943–1977.
- Cuñat V, Giné M, Guadalupe M (2013) Say pays! Shareholder voice and firm performance. ECGI Finance Working Paper 373, Brussels.
- Dowell G, Hart S, Yeung B (2000) Do corporate environmental standards create or destroy market value? *Management Sci.* 46(8):1059–1074.
- Edmans A (2011) Does the stock market fully value intangibles? Employee satisfaction and equity prices. J. Financial Econom. 101(3):621–640.

- Edmans A (2012) The link between job satisfaction and firm value, with implications for corporate social responsibility. *Acad. Management Perspect.* 26(4):1–19.
- Ertimur Y, Ferri F, Stubben S (2010) Board of directors' responsiveness to shareholders: Evidence from shareholder proposals. *J. Corporate Finance* 16(1):53–72.
- Etzion D (2007) Research on organizations and the natural environment, 1992-present: A review. J. Management 33(4):637–664.
- Flammer C (2013) Corporate social responsibility and shareholder reaction: The environmental awareness of investors. Acad. Management J. 56(3):758–781.
- Flammer C (2014) Does product market competition foster corporate social responsibility? Evidence from trade liberalization. *Strategic Management J.*, ePub ahead of print August 22, http://dx.doi.org/10.1002/smj.2307.
- Freeman RE (1984) Strategic Management: A Stakeholder Approach (Pitman, Boston).
- Friedman M (1962) *Capitalism and Freedom* (University of Chicago Press, Chicago, IL).
- Friedman M (1970) The social responsibility of business is to increase its profits. *New York Times Magazine* (September 13), 32–33.
- Gardner Denver (2010) SEC Form DEF 14A. Proxy statement filing, U.S. Securities and Exchange Commission, Washington, DC. Accessed January 31, 2015, http://www.sec.gov/Archives/ edgar/data/916459/000095012310025436/c55699ddef14a.htm.
- Gompers PA, Ishii JL, Metrick A (2003) Corporate governance and equity prices. *Quart. J. Econom.* 118(1):107–155.
- Greening DW, Turban DB (2000) Corporate social performance as a competitive advantage in attracting a quality workforce. *Bus. Soc.* 39(3):254–280.
- Haanaes K, Reeves M, von Streng Velken I, Audretsch M, Kiron D, Kruschwitz N(2012) Sustainability nears a tipping point. *MIT Sloan Management Rev.* 53(2):69–74.
- Hart SL (1995) A natural resource-based view of the firm. Acad. Management Rev. 20(4):986–1014.
- Imbens G, Lemieux T (2008) Regression discontinuity designs: A guide to practice. J. Econometrics 142(2):615–635.
- Jensen M, Meckling W (1976) Theory of the firm: Managerial behavior, agency costs, and capital structure. J. Financial Econom. 3(4): 305–360.
- Jones TM (1995) Instrumental stakeholder theory: A synthesis of ethics and economics. *Acad. Management Rev.* 20(2):404–437.
- Karpoff JM, Malatesta PH, Walkling RA (1996) Corporate governance and shareholder initiatives: Empirical evidence. J. Financial Econom. 42(3):365–395.
- Klassen RD, McLaughlin CP (1996) The impact of environmental management on firm performance. *Management Sci.* 42(8): 1199–1214.
- Lacy P, Cooper T, Hayward R, Neuberger L (2010) A new era of sustainability: UN Global Compact-Accenture CEO Study 2010. Report, Accenture, New York. http://www.accenture .com/sitecollectiondocuments/pdf/accenture_a_new_era_of _sustainability_ceo_study.pdf.
- Leamer EE (1985) Vector autoregressions for causal inference? Carnegie-Rochester Conf. Ser. Public Policy 22(1):255–304.
- Lee DS, Lemieux T (2010) Regression discontinuity designs in economics. J. Econom. Literature 48(2):281–355.
- Lev B, Petrovits C, Radhakrishnan S (2010) Is doing good good for you? How corporate charitable contributions enhance revenue growth. *Strategic Management J.* 31(2):182–200.
- Listokin Y (2008) Management always wins the close ones. Amer. Law Econom. Rev. 10(2):159–184.
- Lohr S (2011) First, make money. Also, do good. *New York Times* (August 13), http://www.nytimes.com/2011/08/14/business/ shared-value-gains-in-corporate-responsibility-efforts.html.

- Loss L, Seligman J (2004) Fundamentals of Securities Regulation, 5th ed. (Aspen Publishers, New York).
- Margolis JD, Walsh JP (2001) People and Profits: The Search for a Link Between a Company's Social and Financial Performance (Lawrence Erlbaum Associates, Mahwah, NJ).
- Margolis JD, Walsh JP (2003) Misery loves companies: Rethinking social initiatives by business. Admin. Sci. Quart. 48(2): 268–305.
- Margolis JD, Elfenbein HA, Walsh JP (2007) Does it pay to be good? A meta-analysis and redirection of research on the relationship between corporate social and financial performance. Working paper, Harvard Business School, Boston.
- McCrary J (2008) Manipulation of the running variable in the regression discontinuity design: A density test. J. Econometrics 142(2):698–714.
- McWilliams A, Siegel D (2001) Corporate social responsibility: A theory of the firm perspective. *Acad. Management Rev.* 26(1):117–127.
- Orlitzky M, Schmidt FL, Rynes SL (2003) Corporate social and financial performance: A meta-analysis. *Organ. Stud.* 24(3): 403–441.
- Peterson DK (2004) The relationship between perceptions of corporate citizenship and organizational commitment. *Bus. Soc.* 43(3):296–319.
- Pfeffer J (1994) Competitive Advantage Through People: Unleashing the Power of the Work Force (Harvard Business School Press, Boston).
- Porter ME (1991) America's green strategy. *Scientific Amer.* 264(4): 96–103.
- Porter ME, Kramer MR (2006) Strategy and society: The link between competitive advantage and corporate social responsibility. *Harvard Bus. Rev.* 84(12):78–92.
- Porter ME, Kramer MR (2011) The big idea: Creating shared value. Harvard Bus. Rev. 89(1-2):62-77.
- Preston LE, O'Bannon DP (1997) The corporate social-financial performance relationship. *Bus. Soc.* 36(4):419–429.
- Reinhardt FL (1998) Environmental product differentiation: Implications for corporate strategy. *Calif. Management Rev.* 40(4): 43–73.
- Roberts MR, Whited T (2013) Endogeneity in empirical corporate finance. Constantinides G, Stulz R, Harris M, eds. Handbook of the Economics of Finance, Vol. 2A (Elsevier, Amsterdam), 493–572.
- Roman R, Hayibor S, Agle BR (1999) The relationship between social and financial performance: Repainting a portrait. Bus. Soc. 38(1):109–125.
- Rusinko CA (2007) Green manufacturing: An evaluation of environmentally sustainable manufacturing practices and their impact on competitive outcomes. *IEEE Trans. Engrg. Management* 54(3):445–454.
- Russo MV, Fouts PA (1997) A resource-based perspective on corporate environmental performance and profitability. Acad. Management J. 40(3):534–559.
- Russo MV, Harrison NS (2005) Organizational design and environmental performance: Clues from the electronics industry. Acad. Management J. 48(4):582–593.
- Turban DB, Greening DW (1996) Corporate social performance and organizational attractiveness to prospective employees. Acad. Management J. 40(3):658–672.
- Vogel D (2005) The Market for Virtue: The Potential and Limits of Corporate Social Responsibility (Brookings Institution Press, Washington, DC).
- Waddock SA, Graves SB (1997) The corporate social performancefinancial performance link. *Strategic Management J.* 28(4): 303–319.